Author's response to reviews

Title: Case Report: Giant Perineal Keloids Treated with Post-Excisional Radiotherapy.

Authors:

Kristin C Jones (joneskc@uthscsa.edu)
Clifton David Fuller (fullercd@uthscsa.edu)
Join Y Luh (iluh@ctrc.net)
Craig C Childs (cchilds@mindspring.com)
Alexander R Miller (amiller@ctrc.net)
Anthony Tolcher (atolcher@ctrc.net)
Terence S Herman (HERMANT@UTHSCSA.EDU)
Charles R Thomas Jr (thomasch@ohsu.edu)

Version: 6 Date: 21 January 2006

Author's response to reviews: see over
To Whom It May Concern:

We are pleased to resubmit for review as our manuscript “Case Report: Giant Perineal Keloids Treated with Post-Excisional Radiotherapy” by Jones et al. All authors have read and approved this manuscript.

We greatly appreciate the opportunity to address the listed concerns of the reviewers and have made all suggested alterations; likewise, we are thankful for the referee’s constructive comments.

Thank you for your consideration of this manuscript,

Sincerely,

CLIFTON D. FULLER, B.S*

KRISTIN JONES, B.S.

*Corresponding author-
Clifton D. Fuller
Department of Radiation Oncology,
7703 Floyd Curl Dr, MSC 7889,
San Antonio, TX, 78229

fullercd@uthscsa.edu
210-616-5648